Author's response to reviews

Title: Low-Grade Pelvic Masses with Spindle Cell and Fibroblastic Proliferation: Case Report

Authors:

John P Micha MD (bram@gynoncology.com)
Bram H Goldstein PhD (bramhg@hotmail.com)
Douglas Ellison MD (dellison@hoaghospital.org)
John V Brown III MD (jebbrown@cox.net)
Mark A Rettenmaier MD (mretten@cox.net)

Version: 2 Date: 20 December 2006

Author’s response to reviews: see over
December 14, 2006

Professor Michael Kidd  
Editor-in-Chief  
Journal of Medical Case Reports  
BioMed Central Ltd  
Middlesex House  
34-42 Cleveland Street  
London W1T 4LB, UK  

RE: manuscript submission

Dear Professor Kidd:

Please find enclosed our manuscript entitled "A Rare Case of Low-Grade Pelvic Masses with Spindle Cell and Fibroblastic Proliferation". We report the status of a patient who presented with large abdominal pelvic masses, ascites, and an elevated serum CA-125 level. Pathology revealed a very rare case of low-grade spindle cell proliferation with fibroblastic features. Following treatment, the patient has since recovered without incident.

Please contact me if I can provide you with any additional information.

Sincerely yours,

John P. Micha, M.D.  
Gynecologic Oncology Associates  
351 Hospital Road, Suite 507  
Newport Beach, CA  92663  
(949) 642-5165